Author’s response to reviews

Title: NAV-KIDS² trial: protocol for a multi-centre, staggered randomised controlled trial of a patient navigator intervention in children with chronic kidney disease

Authors:

Anita van Zwieten (anita.vanzwieten@sydney.edu.au)
Patrina Caldwell (patrina.caldwell@health.nsw.gov.au)
Kirsten Howard (kirsten.howard@sydney.edu.au)
Allison Tong (allison.tong@sydney.edu.au)
Jonathan Craig (jonathan.craig@flinders.edu.au)
Steve Alexander (stephen.alexander@health.nsw.gov.au)
Martin Howell (martin.howell@sydney.edu.au)
Armando Teixeira-Pinto (armando.teixeira-pinto@sydney.edu.au)
Carmel Hawley (Carmel.Hawley@health.qld.gov.au)
Shilpa Jesudason (Shilpa.Jesudason@kidney.org.au)
Amanda Walker (Amanda.Walker@rch.org.au)
Fiona Mackie (fiona.mackie@health.nsw.gov.au)
Sean Kennedy (Sean.Kennedy1@health.nsw.gov.au)
Steve McTaggart (Steven.McTaggart@health.qld.gov.au)
Hugh McCarthy (hugh.mccarthy@health.nsw.gov.au)
Simon Carter (Simon.Carter@health.nsw.gov.au)
Siah Kim (siah.kim@health.nsw.gov.au)
Sam Crafter (Sam.Crafter@sa.gov.au)
Reginald Woodleigh (reg@pbcf.org.au)
Chandana Guha (cguhasyd@gmail.com)
Dear Editorial Office,

We have addressed the concerns raised below:

We notice that most authors are missing from the authors' contributions section. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

We have added the contributions of all authors in page 20.

--  At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

This has been uploaded.

Kind regards

Germaine Wong (on behalf of all authors)